The British Accounting Review xxx (xxxx) xxx



Contents lists available at ScienceDirect

The British Accounting Review



journal homepage: www.elsevier.com/locate/bar

Gender differences in the first course in accounting: An achievement goal approach

Jari Huikku^{a,*}, Emma-Riikka Myllymäki^a, Hannu Ojala^b

^a Aalto University School of Business, Department of Accounting, P.O. Box 21210, 00076, Aalto, Finland
 ^b University of Eastern Finland, Department of Accounting, Aalto University School of Business, P.O. Box 21210, 00076, Aalto, Finland

ARTICLE INFO

Keywords: Achievement goal Gender Accounting education Introductory accounting course Learning environment

ABSTRACT

We examine whether gendered patterns can be observed in first-year students' achievement goals in an introductory accounting course; a question largely overlooked by prior literature. This investigation is motivated by perceptions of accounting as a masculine field involving gender role stereotypes and business schools as competitive and performance-oriented environments. Our findings suggest that male students tend to adopt performance-approach goal, implying that they are more competitive than female students, and that their performance is thus driven by a desire to outperform others. Our findings further suggest that male students' expectations of learning accounting are higher than those of female students. The expectations explain the gender differences in the performance-approach goal. Finally, we find that this performance-approach goal mediates gender differences in course performance depending on the mode of assessment; male students received higher grades for exams but not for teamwork. Overall, our study highlights the importance of considering contextual aspects related to competitiveness, masculinity, and the mode of assessment on an accounting course when addressing students' achievement goals and expectations of learning accounting. We thus contribute to the understanding of how learning environment, accounting pedagogy, and the broader field of professional accounting intersects with individual student attributes, creating differential learning outcomes.

1. Introduction

This paper aims to improve the understanding of potentially gendered differences in students' achievement goals in the context of an introductory accounting course. Prior research on the accounting profession provides some evidence for the existence of gender differences in achievement motivation (Gammie & Whiting, 2013; Whiting & Wright, 2001).¹ Less is known, however, about gender differences in students' achievement goals in accounting.

Achievement goal theory is one of the most influential theories of motivation in education research. It presents two major goal approaches: the *performance-approach* (aiming to outperform others) relates to competitiveness and social comparison, while the *mastery-approach* (which aims to truly comprehend the topic) relates to adaptive motivational processes, including self-efficacy,

* Corresponding author.

E-mail addresses: jari.huikku@aalto.fi (J. Huikku), emma-riikka.myllymaki@aalto.fi (E.-R. Myllymäki), hannu.ojala@uef.fi (H. Ojala).

¹ Whiting and Wright (2001) reported that lower job status and remuneration for women can be partly explained by lower career aspirations. They also demonstrated higher competitiveness in men. Gammie and Whiting (2013) examined why women leave professional accountancy firms and reported that the primary reason was a desire to seek more interesting work.

https://doi.org/10.1016/j.bar.2022.101081

Received 18 December 2019; Received in revised form 8 January 2022; Accepted 12 January 2022

Available online 31 January 2022

0890-8389/© 2022 The Author(s). Published by Elsevier Ltd on behalf of British Accounting Association. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Please cite this article as: Jari Huikku, The British Accounting Review, https://doi.org/10.1016/j.bar.2022.101081

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

intrinsic motivation and interest (e.g., Hulleman et al., 2010; Senko et al., 2011). Prior studies have suggested two additional goals: *performance-avoidance* (the goal of not performing worse than others) and *mastery-avoidance* (the goal of not failing to fully comprehend the topic) (Elliot & Church, 1997; Elliot & McGregor, 2001). The achievement goals are situationally specific orientations (e.g., Harackiewicz et al., 1997); thus, the context matters. It may shape, inflate or mitigate gender differences in goal orientations.

Regarding context, accounting has traditionally been viewed as a masculine discipline (e.g., Bebbington et al., 1997), and stereotypical assumptions about gendered characteristics in accounting may persist (Carmona & Ezzamel, 2016; Haynes, 2017). Gender role stereotypes have the potential to shape females' and males' motivation-related beliefs and behaviours (Meece et al., 2006). Another context-related feature is that studying accounting takes place on business degree programmes, which are generally considered masculine and performance-oriented with an emphasis on exams, grades and career advancement (e.g., Bråten & Strømsø, 2006; Kelan & Jones, 2010; Adcroft, 2011; Lucas & Tan, 2013). Given that men have been found to be more competitive and overconfident and to be more favourably disposed to competition than women (Niederle & Vesterlund, 2007, 2011; Croson & Gneezy, 2009), competitiveness in business schools may affect how men and women approach their studies. The need for new evidence regarding gendered achievement goals in the accounting context is pronounced because the type of motivation that leads to success varies according to the discipline (Breen & Lindsay, 2002).

We set three research questions to guide this study, the first being: *Are there gender differences in student achievement goal orientation in the context of an introductory accounting course*? To further scrutinise gender differences in achievement goals, we consider the role of expectations of learning accounting, which captures motivations and intentions in studying it and the epistemological beliefs about it (Duff & Mladenovic, 2015). Accordingly, our second research question is: *Are gender differences in achievement goal orientation mediated by expectations of learning accounting*? This novel research question pursues a further understanding regarding whether gender affects achievement goals through expectations of learning accounting. Finally, we consider the consequences of gendered goal orientation for course performance. Specifically, we examine whether the gender effect is transmitted through achievement goal orientations to course performance. Therefore, our third research question is: *Are gender differences in course performance mediated by achievement goal orientation*?

The data used in this study consists of first-year undergraduates on a compulsory introductory accounting course in 2017. We utilise survey data on students' achievement goals and expectations of studying accounting, combined with course points, high school grade point average (HSGPA) and student age. Our final sample includes 165 students (105 male students and 60 female students). The achievement goals are measured by latent confirmatory factor analysis constructs based on responses to Elliot and Murayama's (2008) *Achievement Goal Questionnaire-revised* (AGQ-revised). Expectations of learning accounting are measured by factor scores from confirmatory factor analysis based on the main themes studied by Duff and Mladenovic (2015).

The results indicate that male students tend to adopt performance-approach goal. That is, the goal of male students is to outperform others, implying that they are more competitive than female students. The results of performance-approach goal differ from the three other achievement goals with no gender differences. In addition, using covariance-based structural equation modelling, the results of our path analysis indicate that male students' expectations of learning accounting are higher than those of female students, thereby explaining differences in the performance-approach goal. Specifically, we find that gender affects the performance-approach goal directly, and indirectly through expectations of learning accounting.

Finally, we examine course performance and find that men have significantly higher exam points than women. Importantly, employing path analysis, we find that the performance-approach goal mediates the association between gender and course exam points. Accordingly, it is plausible to think that male students' underlying competitiveness as a motivational factor leads to higher exam points. We also use alternative modes of assessment as measures of course performance (teamwork and total points), but then mediation is insignificant. A potential reason for these differences between assessment types is that exams seem to favour the gendered performance-approach goal.

Our paper contributes to the limited body of research on students' achievement goals in accounting and their expectations of studying it. Importantly, and broadly speaking, this paper provides a novel research approach and evidence on the links between gender, goal orientation approaches, learning expectations and performance on an accounting course. As our first contribution, our findings show that having the goal of outperforming other students is evident among male students. This differs from the findings of Bråten and Strømsø (2006) and Tempelaar et al. (2015), who study business students in general (i.e., not focusing on accounting) but find no gender differences in the goal of outperforming others. To the best of our knowledge, only Dull et al. (2015) have explicitly examined achievement goals for accounting, but these scholars do not specifically address potential gender differences in achievement goals.

As our second contribution, we provide evidence of the mediating role of expectations of learning accounting in gender differences regarding achievement goals. Although students' expectations and preconceptions of the subject studied are essential to their learning, they are often overlooked in research (Duff & Mladenovic, 2015). Lucas and Meyer (2005) and Duff and Mladenovic (2015) do indeed investigate gender differences in expectations of learning accounting, but their findings on which gender has higher or lower expectations are overall inconclusive. We extend this stream of research by reporting that higher expectations among male students may explain their performance-approach goal.

Our third contribution is to establish the path from gender to course performance through the performance-approach goal. Our study corroborates the findings of Doran et al. (1991) and Koh and Koh (1999), and, importantly, provides novel evidence that male students' stronger preference for the performance-approach goal explains their better results in examinations but not in teamwork. In other words, our findings suggest that male students' performance is driven by the desire to outperform others and that such competitiveness can lead to higher exam grades. Hence, our study contributes to the existing literature by showing that the role of the achievement goal as a mediator between gender and course performance depends on the mode of assessment.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

Taken together, our paper extends the accounting and education literature by proposing that perceptions of accounting as a masculine field serve to accentuate gender differences in competitiveness, shaping students' performance-approach goals. Consequently, male students' tendency towards performance-approach can be explained by competitiveness and higher expectations of learning accounting. Furthermore, these motivational drivers lead to higher exam grades for male students since exams as a mode of assessment appear to support performance-approach goals.

With regard to practical implications, our study reveals a need to ensure gender equal chances of performing well on introductory accounting courses. Inclusivity is a fundamental issue that merits attention. Learning environment, course content, assessment and teaching methods ought to aim at gender neutrality. Therefore, the status quo needs to be challenged and changes in the learning environment of business schools' accounting courses considered. Although our data on high school GPAs show that female students possess higher levels of prior knowledge, it is a worrying and a clear problem that females perceive accounting as a less attractive subject to study than do male students. We assume that this is due to the perception of accounting as a masculine field (e.g., Bebbington et al., 1997; Carmona & Ezzamel, 2016; Haynes et al., 2017). Our findings lend support to the notion that high-level academic outcomes for both genders would be best ensured by understanding that female students' less competitive goal-setting and lower expectations of learning accounting are aspects clearly needing consideration when designing accounting education. The learning environment should not be biased towards the learning orientation and expectations of any gender. In general, accounting educators should alleviate harmful gender role stereotypes regarding the accounting field and gender differences in competitiveness.

The remainder of the paper is structured as follows. Section 2 discusses the background and research questions. Section 3 describes the data and the model. Section 4 presents the results of the main analyses and discusses the robustness checks and additional analyses. Section 5 discusses our results and concludes the paper.

2. Background and research questions

Achievement goal theory (AGT) has enjoyed considerable standing among theories of motivation in education research since the mid-1980s (see e.g., Senko et al., 2011). Researchers developed AGT to better understand students' motivation and behaviours related to achievement (Dweck, 1986; Nicholls, 1984). The AGT emphasises students' reasons for choosing courses, performing and persevering during their learning activities (Meece et al., 2006). Originally, the AGT consisted of two goal dimensions in a dichotomous framework: a performance-approach goal orientation and a mastery-approach goal orientation (Dweck & Leggett, 1988). A performance-approach goal commonly includes a desire to outperform others, for example in exams (Nicholls, 1984; Hulleman et al., 2010).² Performance is hence linked to ego orientation, relative ability and self-enhancement (Hulleman et al., 2010). The literature has documented that competitiveness is the major predictor of the performance-approach goal (Harackiewicz et al., 1997; Elliot & McGregor, 2001).

Mastery-approach goal-oriented students strive to learn as much as possible; their goal is to truly understand, or master, the task at hand. Accordingly, they focus on developing and improving their skills and competence relative to the task (Harackiewicz et al., 1998; Hulleman et al., 2010). Prior studies have shown that students who adopt the mastery-approach goal have greater motivation characterised by work-mastery (Harackiewicz et al., 1997; Elliot & McGregor, 2001). That is, they desire to work hard and excel, prefer challenging tasks and meet internally prescribed standards of excellence (Elliot & McGregor, 2001).

Subsequently scholars have proposed an extension to the dichotomous framework by adding a performance-avoidance goal (Elliot & Church, 1997; Elliot & Harackiewicz, 1996). Students adopting a performance-avoidance goal try to avoid performing worse than others (Senko et al., 2011). Further, Elliot and McGregor (2001) introduced a mastery-avoidance goal as the fourth category in the achievement goal framework. Elliot and McGregor (2001, p. 502) also give several examples of students demonstrating mastery-avoidance goal behaviour, such as striving to avoid failing to comprehend or learn course material and to retain what they learned. Both mastery-avoidance and performance-avoidance goals are associated with a greater fear of failure and lower self-determination (Elliot & McGregor, 2001).

Prior research has generally found that a mastery-approach goal is associated with positive academic outcomes, whereas the findings on the benefits of the performance-approach goal are less consistent (Hulleman et al., 2010). In some studies, the performance-approach goal has been assumed to have a deleterious effect on a student's academic success, but Harackiewicz et al. (1998), for example, suggest that the performance-approach goal has the potential to enhance students' academic achievement. They state (p. 1) that 'if success does depend on outperforming others, a focus on winning might actually prove adaptive in some college contexts'. Since achievement motivation (i.e., motivation to strive for performance excellence) encompasses features from both mastery and performance-approach goal orientations, Harackiewicz et al. (1997) suggest that achievement-oriented individuals should endorse both orientations. In the context of accounting, Dull et al. (2015) is the only study to examine achievement goals; they conclude that mastery and performance-approach goal motivations jointly lead to better course grades. Regarding avoidance goals (performance-avoidance and mastery-avoidance), it has been suggested that neither has beneficial effects on outcomes (Elliot & Church, 1997; Elliot & McGregor, 2001; Hulleman et al., 2010).

Achievement goals are situationally specific orientations (Harackiewicz et al., 1997), and are therefore somewhat malleable, influenced by the context, and cannot be generalised from one context to another. Therefore, when examining student-specific

 $^{^{2}}$ An appearance component of the performance-approach goal (i.e., the desire to demonstrate superior competence) is sometimes examined separately. In the present study, we follow Elliot and Murayama (2008) and focus exclusively on the normative component of the performance-approach goal, which relates to competition and social comparison.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

characteristics such as gender, it is important to consider the specific context which, in our case, is a first-year accounting course at a business school. Below we will discuss the potentially gendered nature of accounting.

Accounting has traditionally been considered a masculine profession and is characterised as male-dominated (Bebbington et al., 1997); it has even been described as 'universal masculine' or 'absolute masculine' (Hines, 1992, p. 314). The stereotypical assumptions about gendered characteristics in accounting may have diminished over time, but not completely (see the discussion, e.g., by Carmona & Ezzamel, 2016 and Haynes, 2017). Another issue to consider is the common preconception among first-year accounting students, namely, that it is linked to maths, numbers, techniques and formulas (Lucas & Meyer, 2005). Therefore, students often assume that they will struggle on introductory accounting courses if they are not mathematically oriented (Dull et al., 2015; Lucas, 2000). Considering that gender differences in confidence are particularly noticeable in mathematics (Niederle & Vesterlund, 2010), they may also be present in the accounting context.

To view the above discussion in light of achievement motivation, we consider the paper by Meece et al. (2006), who discuss the psychological and education research on gender differences in achievement motivation and conclude that the differences tend to be domain-specific. Importantly, they note that gender role stereotypes shape motivation-related beliefs and behaviours of both genders.³ Therefore, to the extent that gender role stereotypes exist in accounting, this may affect achievement motivation, leading to gender differences in goal orientations.

2.1. Gender and achievement goals

This section will review the literature on the association between gender and achievement goals; we discuss their antecedents and present our first research question. To the best of our knowledge, the literature is silent on an association between achievement goals and gender in the accounting. Below, we summarise what is known about the association at the aggregate level of business schools and in other disciplines.

The performance-approach goal is found to be adopted by male students in many studies (D'Lima et al., 2014; Bouffard et al., 1995; Cavallo et al., 2004). Males have higher levels of overconfidence and competitiveness and are more willing to engage in competitive situations (e.g., Croson & Gneezy, 2009; Niederle & Vesterlund, 2007, 2011).⁴ Accordingly, males are the common denominator in these research findings, suggesting that the performance-approach goal may be gendered.⁵ However, as achievement goals are situationally specific, the greater adoption of performance-approach goal by male students may not be apparent in some disciplines (cf., Harackiewicz et al., 1997; Elliot & McGregor, 2001; Harackiewicz et al., 2002). With regard to business studies, the findings of prior studies are inconclusive. Bråten and Strømsø (2006) found no evidence of gender differences in performance-approach goals among students on a business administration programme, although they did find differences among participants on a teacher training programme. Moreover, using a sample of first-year students studying in a business and economics school, Tempelaar et al. (2015) found no gender differences in the goal to outperform others.⁶

Business studies offer a context commonly characterised as competitive and performance-oriented (e.g., Adcroft, 2011; Bråten & Strømsø, 2006; Lucas & Tan, 2013). According to Bråten and Strømsø (2006), the selection of business studies by competitive and confident females could be a possible reason why gender differences were neutralised. Another explanation could be the differences in the subjects offered within business schools. While the literature reports no clear gender differences at the level of business school, some differences may emerge when we separate the school level into subjects studied. It may be that gender role stereotypes and preconceptions of accounting as masculine (e.g., Bebbington et al., 1997; Carmona & Ezzamel, 2016) accentuate male competitiveness on an accounting course, leading male students to adopt performance-approach goals.

Many studies have reported that female students are likely to adopt mastery-approach goals (D'Lima et al., 2014; Bouffard et al., 1995; Harackiewicz et al., 1997; Elliot & McGregor, 2001; Harackiewicz et al., 2002), although some have found no gender differences (e.g., Cavallo et al., 2004). Importantly, in a business school setting, Bråten and Strømsø (2006) report no evidence, and Tempelaar et al. (2015) report modest (or no) evidence of gender differences in mastery-approach goals.⁷ The accounting context (if perceived as masculine) (Bebbington et al., 1997; Carmona & Ezzamel, 2016), however, may play an important role in explaining gender differences in the mastery approach, which is to a large extent determined by intrinsic motivation to excel in a particular subject. As the

³ Meece et al. (2006) state that important gender differences in educational achievement and occupational attainment still persist although many gaps between females and males have decreased. For example, Meece et al. (2006, p. 367) state that 'boys tend to have positive achievement-related beliefs in the areas of mathematics, science, and sports while girls report more favourable motivation patterns in language, arts and reading'.

⁴ In the literature, the arguments to explain gender differences in competitiveness relate to societal structure (nurture) on the one hand and genetic differences (nature) on the other (see the discussion in Croson & Gneezy, 2009).

⁵ One of the most researched gender differences is the attitude towards risk (which can be related to competitiveness). The research in organisational psychology and economics documents that, on average, women tend to be more risk-averse than men (Croson & Gneezy, 2009). Similar conclusions have been drawn in accounting studies examining, for example, gender diversity of boards of directors (e.g., Dong et al., 2017; Gull et al., 2018). Moreover, Powell and Ansic (1997) showed in a business school environment that females are less risk-seeking than males irrespective of familiarity and framing, costs or ambiguity. In their review of the literature on gender differences, Croson and Gneezy (2009) conclude that potential reasons for women's risk-aversion relate to stronger emotional reactions to risky situations, lower overconfidence and perceiving risks as threats (whereas men tend to perceive them as challenges).

⁶ Tempelaar et al. (2015) utilised Dweck's (1999) inventory, in which normative outcome goals closely corresponded to our measure of the performance-approach goal that captures a student's goal to outperform peers (see Section 3 for details).

⁷ Both these studies used different inventories from ours.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

literature is silent about the gender differences in the adoption of mastery-approach goals in accounting, whether any association exists becomes an empirical question.

The association between avoidance goals and gender has been less studied, and the findings are mixed. When studying first-year students representing various disciplines, D'Lima et al. (2014) found that male students were more performance-avoidance goal-–oriented, whereas in a business school setting, Bråten and Strømsø (2006) reported that females had marginally higher performance-avoidance goal orientation. These scholars did not address the mastery-avoidance goal. In the context of an introductory psychology course, Elliot and McGregor (2001) found no correlation between gender and the adoption of performance-avoidance or mastery-avoidance goals. The accounting literature has not addressed the association between gender and avoidance goals (nor any other achievement goals). Hence, the existence of this association remains another empirical question.

Given the above discussion, we pose the following research question:

RQ1. Are there gender differences in student achievement goal orientation in the context of an introductory accounting course?

2.2. Gender, expectations of learning accounting and achievement goals

In this section, we first discuss the meaning of learning expectations in accounting and the link to achievement goals. Then, we address the potential role of gender in these associations and pose the second research question.

Although scarcely examined, students' expectations are a crucial factor in learning (Lucas & Meyer, 2005; Duff & Mladenovic, 2015).⁸ It has been suggested that many students come to introductory accounting courses with negative perceptions about accounting (see Mladenovic, 2000 for a literature review). This is worrying because these perceptions influence students' learning expectations and can lead to lower performance (Ferreira & Santoso, 2008).

According to Duff and Mladenovic (2015), high and low expectations of learning accounting reflect aspects of motivation (such as personal interest in accounting), intention (regarding learning accounting) and epistemological beliefs (beliefs about what accounting is all about). While the achievement goals seek to explain why students perform and persevere on a course (Meece et al., 2006), expectations capture whether they are interested in performing and persevering in the first place. Hence, higher expectations can drive goal orientations, especially the mastery-approach goal, which emphasises learning and mastering a topic (Hulleman et al., 2010). Surprisingly, and to the best of our knowledge, no studies have examined this important association between expectations of learning accounting and achievement goal orientation.

With regard to gender, the association between gender and expectations of learning accounting has been very little researched, and in the existing studies, the findings appear inconclusive. Lucas and Meyer (2005) examined expectations of learning introductory accounting at UK universities and reported that male students scored higher on lack of interest and had a stronger exam focus. Women scored higher on scales measuring the likelihood of worrying and on the belief that accounting is mainly about numbers and calculations. Differences in the subscales measuring anticipated enjoyment and relevance of accounting proved insignificant between genders.

In a more recent study, Duff and Mladenovic (2015) applied a cluster-analytical approach and divided a sample of first-year undergraduates in Australia into three clusters based on their expectations of learning accounting: in Cluster 1 (moderate expectations), students had an average profile, with scores between Cluster 2 (high expectations) and Cluster 3 (low expectations). The findings suggest no clear gender differences relating to whether females/males had higher expectations regarding learning accounting. Specifically, these authors report that students in the 'moderate expectations' ('low expectations') cluster are marginally more (less) likely to be women. However, they found no gender differences in the 'high expectations' cluster.

Taken together, the above literature presents a somewhat unclear picture of which gender has higher expectations when studying accounting (Lucas & Meyer, 2005; Duff & Mladenovic, 2015). Nevertheless, the perceived masculinity of accounting (Bebbington et al., 1997) may shape expectations of learning accounting, so these expectations may be higher among male students.

By covering a wide range of gender differences in expectations of learning accounting in an introductory accounting course, the research by Lucas and Meyer (2005) and Duff and Mladenovic (2015) has established a solid basis for future studies addressing the role of expectations. We build on their work and examine the unaddressed links between gender, achievement goals and students' expectations of learning accounting. Specifically, considering an association between gender and achievement goals, we examine whether expectations of learning accounting mediate this association and pose the following research question:

RQ2. Are gender differences in achievement goal orientation mediated by expectations of learning accounting?

2.3. Gender, achievement goals and course performance

To examine the consequences of potential gender differences in achievement goals, we explore course performance. The research so far provides mixed evidence on whether there are gender differences in terms of performance on introductory accounting courses (e.g., in exam grades). Some studies report insignificant differences between genders (e.g., Buckless et al., 1991; Carpenter et al., 1993; Duff, 2004; Gist et al., 1996; Lipe, 1989), whereas others report that female students outperform male students (e.g., Arthur & Everaert, 2012; Crawford & Wang, 2014; Tho, 1994; Tyson, 1989). There are also studies suggesting that male students perform significantly

⁸ Also, in other disciplines, such as in mathematics (Meyer & Eley, 1999) and economics (Meyer & Shanahan, 2001), subject-specific conceptions have been identified as playing a major role in students' motivations to learn (see also Breen & Lindsay, 2002).

J. Huikku et al.

better than female students (e.g., Doran et al., 1991; Koh & Koh, 1999).

Assessment methods (such as project work, presentations and exams) are essential components of course learning environments since they convey information about what is valued, hence influencing students' learning (Struyven et al., 2005). Students may adapt themselves to the assessment methods in use, leading to different course performance from what the performance would have been had another assessment method been used. Next, we discuss the potential association between gender, achievement goals and course performance and consider the role of different modes of assessment.

Some studies have found that male students have more positive attitudes towards exams as an assessment method (e.g., Furnham & Chamorro-Premuzic, 2005), whereas others have not found such evidence (e.g., Turner & Gibbs, 2010; Woodfield et al., 2005). Regarding achievement goals, studies have suggested a positive association between the performance-approach goal and exam performance (Elliot & McGregor, 2001; Harackiewicz et al., 1997). For example, Harackiewicz et al. (1997) discuss how in introductory psychology courses, the large lecture format, multiple-choice exams and a normative grading structure potentially establish a context in which the performance-approach goal is conducive to course performance. Overall, it may be that the competitiveness of male students is linked to a greater performance goal orientation (see Section 2.1.), which in turn may lead to higher points on exams (i.e., an assessment method that favours a performance orientation).

Compared to male students, females have been found in some studies to have more positive attitudes towards continuous assessment, including assignments and projects, with no rigorous exam situations (e.g., Furnham & Chamorro-Premuzic, 2005). Yet Woodfield et al. (2005) and Turner and Gibbs (2010), among others, found no gender differences. Regarding overall course grades, Dull et al. (2015) found that female students performed better on an accounting course consisting of projects, quizzes, exams and homework assignments, but that there were no gender differences in the exam grades. Gammie et al. (2003) found no gender differences in course performance when assessed on the basis of course work or exams among final-year accounting and finance students. Collaborative learning seems to be preferred by females (e.g., Curşeu et al., 2018) and strengthened by the mastery-approach goal (e.g., Hijzen et al., 2007; Senko et al., 2011). Also, in the collaborative learning context, Senko et al. (2011) point out that the underlying competitiveness of the performance approach may be detrimental to the team members' learning and may affect overall group performance.

Overall, in light of the above discussion, achievement goals may mediate the effect of gender on course performance if the assessment methods support gendered goal orientations. This aspect has been overlooked in the literature. We add to the literature by examining whether gender differences in course performance can be explained by potentially gendered achievement goals. In so doing, we also test whether assessment methods (exams, teamwork and total points) can play a role. Accordingly, we pose the following research question:

RQ3. Are gender differences in course performance mediated by achievement goal orientation?

3. Data and model

3.1. Data and participants

The data used in this study were obtained from first-year undergraduates taking a compulsory introductory accounting course at a university in Finland. Finland is an appropriate research setting for the purposes of this study because in the Nordic countries, gender equality is more advanced than elsewhere (Looze et al., 2018, p. 1079), thus helping to capture gender-specific voluntary choices and hence diminishing potential cultural impacts that may bias the findings.⁹

Our primary data were collected from students who joined the course in 2017. Similar to the majority of introductory courses in business schools around the world, this course was a mass course with more than three hundred students. The course had two instructors, one woman and one man, which mitigated the potential interactive effect of (fe)male students in (fe)male-instructed classes (e.g., Lipe, 1989). The course was offered during the last half of the first semester. Hence, by the time the course started, the first-year students had had only two months of studies in this undergraduate programme.

The data were obtained from three sources. First, a (survey) questionnaire elicited information about the students' achievement goals on the introductory accounting course and their expectations of learning accounting.¹⁰ The survey was administered during the first lecture of the course by two of the authors. Second, university entrance data were accessed to gather information about the students' school grades (based on a standard national examination). Third, we used data on the students' course performance. We addressed the ethical aspects by acquiring permission from the university and from the responding students to use the data.

The initial sample size was 321, representing all the students who registered for the course (38% of whom were female). Then, we excluded students who were not studying business (e.g., students studying in other schools of the university) and those who were not first-year students, which left us with 264 student observations. After we excluded those students who did not take the first exam of the course (leaving 229) and also those who did not answer the survey questions about achievement goals or expectations of learning accounting, the final sample size is 165 student observations (105 male students [64%] and 60 female students [36%]).

⁹ However, we acknowledge that there are factors such as social support systems (i.e., family, peers and educational context) that are important from the gender equality point of view (Looze et al., 2018).

¹⁰ To obtain a high response rate, the questionnaire was designed to be short (but comprehensive). To transfer all the data into an analysable form (e.g., transferring the questionnaire data to Excel), one researcher did the coding and another non-coding researcher double-checked for accuracy.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

The additional analysis of course performance uses data consisting of students' course grades in the introductory accounting course between 2012 and 2017. That sample size is 1702 observations, among whom 1086 (64%) are male and 616 (36%) are female. This sample is used in our univariate tests to investigate potential gender differences in course performance across the six years.

3.2. Research design

3.2.1. Structural equation model

In our main analyses, we employ covariance-based structural equation modelling to examine the three research question operationalisations, which are presented in Fig. 1. That is, the figure illustrates the overall picture of the associations addressing the research questions. Our analysis proceeds in three steps. First, for RQ1, we examine the direct association between gender and achievement goals. Then, for RQ2 and RQ3, we employ path models where some variables influence others either directly or through intervening or mediator variables (Preacher & Hayes, 2008). Structural equation modelling enables us to present how constructs are theoretically linked and test the directionality of the significant relationships (Schreiber et al., 2006). We apply a two-phase approach as proposed by Mueller and Hancock (2008); that is, we first conduct a measurement phase (using confirmatory factor analysis), followed by a structural phase. The statistical analysis used 165 observations, which exceeded the minimum of 150 observations that should be used in structural equation modelling (Anderson & Gerbing, 1988, p. 415).

3.2.2. Variables

The variables used in the analyses are presented in Fig. 1 with measured variables represented as rectangles and latent variables as ellipses. *FEMALE* is the variable affecting achievement goals directly (arrow *a* in Fig. 1; RQ1). *FEMALE* is an indicator variable with a value of 1 when the student is female and otherwise zero. Achievement goals consist of four latent goal measures constructed using confirmatory factor analysis. *FEMALE* also affects achievement goals indirectly through *HIGH* (*LOW*) *EXPECTATIONS* (arrows *b* and *c* in Fig. 1; RQ2). *HIGH* (*LOW*) *EXPECTATIONS* are latent constructs from confirmatory factor analysis. *FEMALE* also affects course performance directly (arrow *d* in Fig. 1) and indirectly through achievement goals (arrows *a* and *e* in Fig. 1; RQ3). We measure three assessment types of course performance: the course exam (*EXAM POINTS*), teamwork points (*TEAMWORK*)¹¹ and total course points (*TOTAL POINTS*).¹²

In our analysis, we control for student knowledge. The first control variable measures high school grade point average (*PRIOR KNOWLEDGE*) (e.g., Rankin et al., 2003). Including this measure is relevant to our research setting because there is evidence of a positive association between high school grade point average and achievement at university (Betts & Morell, 1999). The second control variable, student age (*AGE*), captures the experience and is a likely determinant of preferences, attitudes towards studying and self-regulation (e.g., Rankin et al., 2003). See Appendix for variable definitions.

3.2.3. Confirmatory factor analysis

To construct achievement goal variables to examine RQ1, RQ2 and RQ3, we employ the AGQ-revised (Elliot & Murayama, 2008), which has been used extensively (e.g., Cook et al., 2017; Hall et al., 2016) in prior studies. In the questionnaire, the students reported the extent to which they agreed with each item on a five-point scale ranging from totally disagree (1) to totally agree (5).¹³ The questionnaire included 12 questions that underpin the four latent constructs: *PERFORMANCE-APPROACH*, *MASTERY-APPROACH*, *PERFORMANCE-AVOIDANCE* and *MASTERY-AVOIDANCE*.

Table 1 presents the latent constructs, items that measure the latent constructs, standardised factor loadings, the decision criterion to retain the item in the scale of the factor or to delete it from the scale and, finally, the Cronbach's alpha. In Table 1, the standardised factor loadings can be interpreted roughly as a correlation metric (Furr, 2011), and they reflect the degree to which each indicator is linked to a corresponding achievement goal. The standardised loadings show that the items do indeed load very well on the four achievement goals. As seen in the table, the minimum standardised loading of all three indicators of *PERFORMANCE-APPROACH* is 0.916. The indicator 'I am striving to avoid an incomplete understanding of the course material' of *MASTERY-AVOIDANCE* has the weakest standardised loading at 0.595. None of the items have a standardised factor loading below 0.4. In light of this property, we retained all the items in the scales measuring the achievement goals.

From Table 1, we can also see that all subscale scores demonstrate adequate internal consistency. For PERFORMANCE-APPROACH, MASTERY-APPROACH, PERFORMANCE-AVOIDANCE and MASTERY-AVOIDANCE, the respective Cronbach's alpha values are 0.947, 0.745, 0.880 and 0.718. The commonly cited benchmark of the subscale scores provided by Nunnally (1978) suggests that Cronbach's alpha values between 0.7 and 0.8 are satisfactory, 0.8 and 0.9 good, and above 0.9 excellent. Hence, the Cronbach's alpha values in our analysis range from satisfactory (MASTERY-APPROACH and MASTERY-AVOIDANCE) to excellent (PERFORMANCE-APPROACH).

To examine RQ2, we employ nine indicators adapted from Duff and Mladenovic (2015), who show (in the correlation matrix) that there are two sets of expectations: high and low (p. 329). Accordingly, we construct a high expectation factor from the first five items and a low expectation factor from the last four items. In the questionnaire, students reported the extent to which they agreed with each

¹¹ The students were randomly assigned to teams at the beginning of the course. The total teamwork points consist of teamwork (max. 15 points), participation in class exercises to discuss their teamwork (max. 6 points) and an individual reflection on the teamwork (max 5 points).

¹² The total assessment of the course performance (max. 106 points) comprises the final course exam (max. 60 points), two open-book mid-term exams (max. 20 points) and teamwork points (max. 26 points).

¹³ We replaced eight missing item values with the sample median of each item.





Fig. 1. The associations under examination.

Table 1

Construct assessment regarding achievement goals (n = 165).

Construct	Indicator	Loading	Criterion	Decision	Cronbach's Alpha ^{a)}
Performance-approach					0.947
	I am striving to do well compared to other students.	0.924	≥ 0.40	Retained	
	My aim is to perform well relative to other students.	0.916	≥ 0.40	Retained	
	My goal is to perform better than the other students	0.936	≥ 0.40	Retained	
Mastery-approach					0.745
	My aim is to completely master the material presented in this class.	0.673	≥ 0.40	Retained	
	My goal is to learn as much as possible.	0.658	≥ 0.40	Retained	
	I am striving to understand the content of this course as thoroughly as possible.	0.838	≥ 0.40	Retained	
Performance- avoidance					0.880
	My goal is to avoid performing poorly compared to others.	0.741	≥ 0.40	Retained	
	I am striving to avoid performing worse than others.	0.980	≥ 0.40	Retained	
	My aim is to avoid doing worse than other students.	0.822	≥ 0.40	Retained	
Mastery-avoidance					0.718
	My aim is to avoid learning less than I possibly could.	0.903	≥ 0.40	Retained	
	My goal is to avoid learning less than it is possible to learn.	0.629	≥ 0.40	Retained	
	I am striving to avoid an incomplete understanding of the course material.	0.595	≥ 0.40	Retained	

^a Cronbach's alpha is presented based on the retained indicators, here with all indicators, as they exceed the loading threshold of 0.40.

item on a Likert-scale from totally disagree (1) to totally agree (5).¹⁴

In Table 2, we can see that seven out of nine standardised coefficients of the indicators exceed the threshold of 0.40. Regarding the *HIGH EXPECTATIONS* construct, the indicator 'I would be interested in exploring the social and economic importance of accounting'

¹⁴ We replace one missing item value with the sample median.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

has the highest loading of 0.752. The lowest loading, 'I expect that I might question the basis on which accounting techniques are founded', loaded poorly (standardised loading of 0.172), not exceeding the threshold of 0.40, and is therefore deleted. After that, four indicators remain for *HIGH EXPECTATIONS*, together yielding a Cronbach's alpha of 0.722.

The indicators of *LOW EXPECTATIONS* do not load as well as those of *HIGH EXPECTATIONS*. The indicator 'I do not have a personal interest in accounting, and I expect it to be boring' has the highest standardised loading of 0.654. One of the indicators, 'I expect that studying accounting focuses on numbers', has a standardised loading (of 0.209) that does not exceed the threshold of being retained as an indicator for *LOW EXPECTATIONS* and is therefore deleted. After that, three indicators remain for *LOW EXPECTATIONS* that yield a Cronbach's alpha of 0.543, which is below 0.6, an often-used cut-off value for the scale reliability. Because of low scale reliability, we will use caution when interpreting the results regarding *LOW EXPECTATIONS* and will focus on interpreting the results regarding *HIGH EXPECTATIONS*.

4. Results

4.1. Descriptive statistics

Table 3 presents descriptive statistics (mean, median and standard deviation) of the variables used in the analyses for women (60 students) and men (105 students). It also presents the results of differences between the genders in means (*t*-tests) and medians (*z* values from Wilcoxon–Mann–Whitney).

The descriptive statistics show that female students have lower expectations (*LOW EXPECTATIONS*) and better high school grades (*PRIOR KNOWLEDGE*) than male students. The mean factor score of *LOW EXPECTATIONS* for females is 0.058 and 0.033 for male students. The mean factor score of *PRIOR KNOWLEDGE* for females is 29.7 and 26.4 for male students. Female students score lower in the performance-approach goal (*PERFORMANCE APPROACH*), and high expectations (*HIGH EXPECTATIONS*) and have lower exam and total course points (*EXAM POINTS* and *TOTAL POINTS*) than male students. The respective means for females and males are -0.393 versus 0.255; -0.064 versus 0.036; 41.4 versus 44.8; and 80.0 versus 84.1. The differences in medians are otherwise similar, but *AGE* differs between genders. The median age of females (males) is 20 (21) years.

Spearman's correlations (untabulated) indicate that *FEMALE* is negatively correlated with *PERFORMANCE-APPROACH*, *EXAM POINTS*, *TOTAL POINTS* and *AGE*. *FEMALE* is positively correlated with LOW EXPECTATIONS and *PRIOR KNOWLEDGE*.

4.2. Main analyses

The results of examining the three research questions are presented in Figs. 2–4. In Fig. 2, we report the results of the direct association between gender and achievement goals (RQ1). In Figs. 3 and 4, we report the results estimated by the structural equation model (RQ2 and RQ3). The (untabulated) fit statistics of the structural model indicate a good fit. χ 2 is statistically insignificant (p = 0.173), suggesting that there are no unidentified paths, CFI and TLI are close to 1 (0.992 and 0.979 respectively), RMSEA is 0.045, the standardised root mean squared residual (SRMR) 0.062, and the coefficient of determination 0.236.

4.2.1. Results for RQ1

We start from the testing of the association of RQ1 and report the results in Fig. 2. We can see that *FEMALE* is significantly associated with *PERFORMANCE-APPROACH* (coefficient -0.576, p-value <0.001), but not with the three other achievement goals.¹⁵ Hence, we obtain support for gender differences only in performance-approach goal orientation in the context of an introductory accounting course, but not in the other (three) goal orientations.

4.2.2. Results of mediation effects

To examine mediation effects (RQ2 and RQ3), we use the widely cited guidelines for assessing mediation by Baron and Kenny (1986). The testing of causal steps has been provided for RQ2 (Panels A and B) and RQ3 (Panels C through E) in Table 4. The testing includes three steps (three regressions) for each potential mediation path. According to Baron and Kenny (1986), all three sub-paths must be statistically significant as a prerequisite of a mediation test. The last column of the table shows the decision based on the three steps. The decision informs whether the path is retained for the actual mediation test or whether it is deleted from further analyses (as the steps failed to indicate that there is sufficient basis for the mediation test).

4.2.2.1. Results for RQ2. The first potential mediation based on RQ2 is presented in Panel A: the path *FEMALE* \rightarrow *HIGH EXPECTATIONS* \rightarrow *PERFORMANCE-APPROACH*. We require that (1) *FEMALE* has to be a significant predictor of *PERFORMANCE-APPROACH*, (2) *FEMALE* has to be a significant predictor of *HIGH EXPECTATIONS* and (3) *HIGH EXPECTATIONS* has to be a significant predictor of *PERFORMANCE-APPROACH*. Since all three significant associations existed, the decision is to retain the mediation path for the actual mediation test. Similarly, the path *FEMALE* \rightarrow *LOW EXPECTATIONS* \rightarrow *PERFORMANCE-APPROACH* is significant (Table 4, Panel B) and is retained for the actual mediation test. However, since *FEMALE* is not associated with *MASTERY-APPROACH*, *PERFORMANCE-APPROACH*, *PERFOR*

¹⁵ The explanatory power of the model where *FEMALE* and control variables (*AGE* and *PRIOR KNOWLEDGE*) are predictors of *PERFORMANCE-APPROACH* is 11.4%.

J. Huikku et al.

Table 2

Construct assessment regarding expectations of accounting (n = 165).

Construct	Indicator	Loading	Criterion	Decision	Loading ^{a)}	Cronbach's Alpha ^{b)}
High Expectations						0.722
	I have a strong desire to excel in my academic achievement	0.464	≥ 0.40	Retained	0.468	
	I expect that I will enjoy accounting studies.	0.684	≥ 0.40	Retained	0.676	
	I expect that I might question the basis on which accounting techniques are founded.	0.172	< 0.40	Deleted		
	I want to see the meaning behind accounting numbers in a business context.	0.633	≥ 0.40	Retained	0.642	
	I would be interested in exploring the social and economic importance of accounting	0.752	≥ 0.40	Retained	0.749	
Low Expectations						0.543
	I expect that I will do only enough work to simply pass the exam.	0.428	≥ 0.40	Retained	0.424	
	I do not have personal interest in accounting, and I expect it to be boring.	0.654	≥ 0.40	Retained	0.673	
	I expect that studying accounting focuses on numbers.	0.209	< 0.40	Deleted		
	I am worried about my learning in accounting.	0.538	≥ 0.40	Retained	0.521	

^a We re-estimated standardised factor loading after deleting two indicators with below 0.40 standardised loading.

^b Cronbach's alpha is presented based on the retained indicators.

Table 3

Descriptive statistics (n = 165).

	Female (n = 60)		Male (n = 105)			Difference in Means			
	Mean	Median	Std.dev.	Mean	Median	Std.dev.	Difference	t-test	Mann-Whitney
FEMALE	1.000	1.000	0.000	0.000	0.000	0.000	1.000		
PERFORMANCE- APPROACH	-0.393	-0.418	0.996	0.225	0.212	0.819	-0.618	< 0.001***	<0.001***
MASTERY- APPROACH	-0.020	-0.051	0.484	0.011	-0.051	0.467	-0.031	0.687	0.864
PERFORMANCE- AVOIDANCE	-0.066	-0.053	0.844	0.038	-0.053	0.695	-0.104	0.398	0.758
MASTERY- AVOIDANCE	0.050	-0.106	0.785	-0.029	0.030	0.824	0.079	0.548	0.855
HIGH EXPECTATIONS	-0.064	-0.051	0.360	0.036	0.068	0.291	-0.100	0.053*	0.091*
LOW EXPECTATIONS	0.058	0.059	0.269	-0.033	-0.064	0.193	0.091	0.013**	0.042**
EXAM POINTS	41.400	43.000	8.166	44.781	45.000	6.906	-3.381	0.005***	0.008***
TEAMWORK	12.817	13.000	1.501	12.581	13.000	1.610	0.236	0.355	0.387
TOTAL POINTS	80.000	81.000	10.772	84.133	87.000	9.696	-4.133	0.013**	0.007***
AGE	20.317	20.000	2.418	20.790	21.000	1.885	-0.473	0.164	<0.001***
PRIOR KNOWLEDGE	29.700	33.000	8.100	26.429	29.000	9.355	3.271	0.025**	0.015**

* shows significance at the 0.10 level, ** shows significance at the 0.05 level, and *** shows significance at the 0.01 level.

AVOIDANCE and MASTERY-AVOIDANCE (Fig. 2), their corresponding mediation paths are deleted from the actual mediation test. Fig. 3 presents the results for RQ2. The mediation effect is defined as the product of two sub-paths: *FEMALE* to *HIGH* (or *LOW*) *EXPECTATIONS* and *HIGH* (or *LOW*) *EXPECTATIONS* to achievement goal (see Brown, 1997 and Fox, 1985 for testing of paths coefficients). The path from *FEMALE* to *HIGH EXPECTATIONS* is negative and significant (coefficient is -0.100 and p-value 0.05). The path from *HIGH EXPECTATIONS* to *PERFORMANCE-APPROACH* is positive and significant (coefficient is 1.318 and p-value <0.001). The path from *FEMALE* to *PERFORMANCE-APPROACH* through *HIGH EXPECTATIONS* has a z-value of 1.88, with a (two-sided) p-value of 0.06. This lends support to RQ2 with regard to the path from *FEMALE* to *PERFORMANCE-APPROACH* through *HIGH EXPECTATIONS* is insignificant.

4.2.2.2. Results for RQ3. Next, we turn to the empirical testing of RQ3. In Panels C to E of Table 4, we examine only those paths that involve PERFORMANCE-APPROACH, since FEMALE is not associated with the other three goals. Table 4 presents the following three paths: FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow EXAM POINTS (Panel C), FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow TEAMWORK (Panel D), and FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow TOTAL POINTS (Panel E). The paths in Panels C and E are retained for the actual mediation test, but the path presented in Panel D is excluded because TEAMWORK is neither statistically significantly associated with FEMALE nor PERFORMANCE-APPROACH.

The results of the mediation test of RQ3 are presented in Fig. 4, which shows that the sub-path from *FEMALE* to *EXAM POINTS* is negative and significant (coefficient is -2.597 and p-value 0.029). The sub-path from *FEMALE* to *PERFORMANCE-APPROACH* is negative and significant (coefficient is -0.486 and p-value <0.001). The sub-path from *PERFORMANCE-APPROACH* to *EXAM* is positive and significant (coefficient is 1.552 and p-value 0.019). The mediation test shows that the path from *FEMALE* to *EXAM POINTS* through *PERFORMANCE-APPROACH* has a z-value of 2.00, with a (two-sided) p-value of 0.045. These results lend support to RQ3, where the measure of course performance is exam points. The path from *FEMALE* to *TOTAL POINTS* through *PERFORMANCE-APPROACH* is statistically insignificant with a z-value of 1.47 and (two-sided) p-value of 0.141.



Fig. 2. Testing of RO1.

4.3. Robustness tests

We conduct various robustness tests. Regarding RQ1, we run an untabulated one-way MANOVA to determine the effect of gender on achievement goals and include *HIGH EXPECTATIONS, LOW EXPECTATIONS, AGE* and *PRIOR KNOWLEDGE* as covariates. The results of the MANOVA indicate a statistically significant difference in the combined achievement goals between female students and male students (F = 7.28, p = <0.001). A post hoc analysis shows that there are gender differences only in *PERFORMANCE-APPROACH* (significant at a 5% confidence level).

To further explore the association between *FEMALE* and *PERFORMANCE-APPROACH*, we test the gender differences using subsamples of below (above) median positive (negative) expectations. Males have a positive and statistically significant association with the performance goal approach in all four subsamples, regardless of higher or lower expectations. This suggests that the association between gender and performance-approach is strong.

Regarding RQ3, given that earlier research has suggested that men perform better than women when presented with multiplechoice questions (e.g., Arthur & Everaert, 2012; Murphy, 1982; Walstad & Robson, 1997), we test the robustness of our results when we exclude the points of multiple-choice questions (maximum eight points) from the *EXAM POINTS* variable. The findings are qualitatively similar to those from our main analysis (results not tabulated).

Also, to test the broader generalisability of the results for RQ3, we utilise a sample of students' course grades on the introductory accounting course between 2012 and 2017.¹⁶ Although we do not have survey data on achievement goals and expectations of learning accounting for those six years, investigating the observed gender difference in course performance over a longer period could yield further insights into the main findings. Here, course performance is measured by the final grade from the course, which ranged from 1 to 5 (5 being the top grade). The untabulated results indicate statistically significant differences between female and male students. On average, male students have higher grades than females in each year investigated, although the magnitude of the differences varies slightly. The differences are statistically significant at the 0.01 level each year, except for 2014 and 2017, when the difference is significant at the 0.05 level. The test results derived from the total sample (all years) show that the average grade for female students is

¹⁶ The sample includes all students who participated in the course (i.e., first-year business students, later-year business students and those not studying business).



Fig. 3. Testing of RQ2.

3.55, while for male students it is 3.97. In summary, the gender difference in course performance persists over the six-year period, which encompasses all the data available in this study.

Finally, we perform an untabulated discriminant analysis with gender as the grouping variable and other variables as independent variables. The results are largely consistent with the empirical analyses using SEM. The highest standardised coefficient is *PERFOR-MANCE-APPROACH* (coefficient is -0.721). The standardised coefficient of *EXAM* is negative (coefficient is -0.411), the standardised coefficient of *PRIOR KNOWLEDGE* is positive (coefficient is 0.420), and the standardised coefficient of *HIGH EXPECTATIONS* is positive but relatively low (coefficient is 0.062). In the discriminant analyses, we remove variables that are largely redundant because redundancy can lead to unreliable matrix inversions, which result in large standard errors of the estimates of discriminate analysis. We also remove *LOW EXPECTATIONS* because of its high correlation with *HIGH EXPECTATIONS*. The reason for the removal is that when predictor variables are correlated, the associated discriminating coefficients may yield misleading results (Cui, 2012). We decided to remove *LOW EXPECTATIONS* rather than *HIGH EXPECTATIONS* because of the former's low Cronbach's alpha. We use 0.30 in the canonical structure as the redundancy criterion and remove *TEAMWORK* and *AGE*. The discriminant function for *FEMALE* is significant (p-value <0.001).

5. Discussion and conclusion

This study addresses the rarely researched topics of achievement goals for the accounting course and expectations of learning accounting, focusing on the role of gender. The importance of studying motivation-related gender effects has been accentuated by studies investigating gender differences in competitiveness (Croson & Gneezy, 2009), the perception of accounting as a masculine field, gender-role stereotypes (Bebbington et al., 1997; Carmona & Ezzamel, 2016; Haynes, 2017) and gender differences in career aspirations in accounting (e.g., Whiting & Wright, 2001). Consequently, in this paper, we maintain that it is essential to examine the gender effects in achievement goals, high and low expectations of learning accounting, and course performance in the first year of university studies in a business school. In addressing potentially gendered achievement goals and expectations, we can contribute to the accounting and education literature in many ways.

As the first contribution, this is the first study to scrutinise gender differences in students' achievement goals in the accounting context. To the best of our knowledge, Dull et al. (2015) is the only study to examine achievement-goal aspects in the accounting education context, but that study did not address gender differences in achievement goals. Bråten and Strømsø (2006) and Tempelaar et al. (2015) studied gender differences in achievement goals among business students generally, but they did not focus on accounting.

The British Accounting Review xxx (xxxx) xxx



The path $FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow EXAM POINTS$ has a z-value of 2.00 and a (two-sided) p-value of 0.045.

The path $FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow TOTAL POINTS$ has a z-value of 1.47 and a (two-sided) p-value of 0.141.

Fig. 4. Testing of RQ3.

We enrich their findings by providing clear evidence that male students have a stronger performance-approach goal orientation than female students, whereas no gender differences were found in other achievement goals. More specifically, our findings show that among the students surveyed, the goal to outperform others is evident among men, implying that greater competitiveness drives male students' achievement goals.

We interpret that these gender differences in the performance-approach goal are driven by the context characterised by competitiveness (e.g., Bråten & Strømsø, 2006; Kelan & Jones, 2010; Adcroft, 2011; Lucas & Tan, 2013) and, importantly, perceptions of masculinity in accounting (Bebbington et al., 1997). Bråten and Strømsø (2006) and Tempelaar et al. (2015) found no gender differences among business school students in the goal to outperform others. A plausible explanation for this failure to find gender differences is their research setting, where various business school subjects were examined at the aggregate level without addressing accounting separately. We conjecture that men's goal to outperform others is accentuated when the context involves competitiveness, especially in a subject of study commonly characterised as masculine, such as accounting. That is, when the nature of a field is perceived as masculine, it may create more incentive to compete. It is worth noting that we found no gender differences in mastery-approach (the desired goal in higher education) or in avoidance goals (less-desired goals).

As the second contribution, we extend the achievement goal literature by being the first to examine the association between expectations of learning accounting and achievement goals and the mediating role of expectations in the gender differences in achievement goals. We provide evidence to show how gender differences in the performance-approach goal can be attributed to gender differences in expectations of learning accounting. This extends the literature addressing the link between gender and students' expectations of learning accounting (Duff & Mladenovic, 2015; Lucas & Meyer, 2005).

Specifically, we found that male students' expectations of learning accounting are significantly higher than those of female students. Further, the results of the path analysis suggest that the higher expectations of male students explain (mediate) their performance-approach goals. As a whole, we assume that the perceived masculinity of accounting (Bebbington et al., 1997) and competitiveness may explain the higher expectations of learning accounting among male students, which leads to the performance-approach goal.

Our third contribution stems from establishing a path from gender to course performance through the performance-approach goal. Accounting education studies have not addressed this path. Our study corroborates the findings of Doran et al. (1991) and Koh and Koh (1999) reporting that male students obtain better exam points than female students on an introductory accounting course. Importantly, we extend these studies by providing evidence that male students' greater preference for a performance-approach goal explains

The British Accounting Review xxx (xxxx) xxx

Table 4

Testing of mediation steps (n = 165).

Panel A. RQ2:	Path: FEMALE \rightarrow HIGH EXPECTATIONS \rightarrow	PERFORMANCE-APPROACH						
Subpath	Predictor	Dependent	Coefficient	p-value	Decision			
					Retained			
а	FEMALE	PERFORMANCE-APPROACH	-0.576	< 0.001				
b	FEMALE	HIGH EXPECTATIONS	-0.100	0.050				
c	HIGH EXPECTATIONS	PERFORMANCE-APPROACH	1.394	< 0.001				
Panel B. RQ2:	Path: FEMALE \rightarrow LOW EXPECTATIONS \rightarrow	PERFORMANCE-APPROACH						
Subpath	Predictor	Dependent	Coefficient	p-value	Decision			
					Retained			
а	FEMALE	PERFORMANCE-APPROACH	-0.576	< 0.001				
b	FEMALE	LOW EXPECTATIONS	0.901	0.012				
c	LOW EXPECTATIONS	PERFORMANCE-APPROACH	-1.505	< 0.001				
Panel C. RQ3:	Panel C. RO3: Path: FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow EXAM POINTS							
Subpath	Predictor	Dependent	Coefficient	p-value	Decision			
					Retained			
d	FEMALE	EXAM POINTS	-3.444	0.004				
а	FEMALE	PERFORMANCE-APPROACH	-0.576	< 0.001				
e	PERFORMANCE APPROACH	EXAM POINTS	2.568	< 0.001				
Panel D. RQ3:	Path: FEMALE \rightarrow PERFORMANCE-APPROA	$ACH \rightarrow TEAMWORK$						
Subpath	Predictor	Dependent	Coefficient	p-value	Decision			
					Deleted			
d	FEMALE	TEAMWORK	0.215	0.404				
а	FEMALE	PERFORMANCE-APPROACH	-0.576	< 0.001				
e	PERFORMANCE APPROACH	TEAMWORK	0.188	0.153				
Panel E. RQ3: Path: FEMALE \rightarrow PERFORMANCE-APPROACH \rightarrow TOTAL POINTS								
Subpath	Predictor	Dependent	Coefficient	p-value	Decision			
					Retained			
d	FEMALE	TOTAL POINTS	-4.542	0.005				
а	FEMALE	PERFORMANCE-APPROACH	-0.576	< 0.001				
e	PERFORMANCE APPROACH	TOTAL POINTS	3.071	< 0.001				

Table 4 illustrates the results of our examination of the three steps that, according to Baron & Kenny (1986), establish mediation. The last column of the table shows the decision based on the three steps, whether the path was retained for actual mediation test or whether it was deleted from the further analysis, as the steps failed to indicate that there was sufficient basis for mediation test. The first column on the left shows the subpath (as presented in Fig. 1) under examination. According to Baron & Kenny (1986), all of the three subpaths have to be statistically significant as a prerequisite of a mediation test. For example, the first potential mediation based on RO2 is presented in Panel A: the path FEMALE = > HIGH EX-PECTATIONS = > PERFORMANCE-APPROACH. Baron & Kenny (1986) require that (1) FEMALE has to be a significant predictor of PERFORMANCE-APPROACH, (2) FEMALE has to be a significant predictor of HIGH EXPECTATIONS and (3) HIGH EXPECTATIONS has to be a significant predictor of PERFORMANCE-APPROACH. If all of the three significant associations exist, the decision is to retain the mediation path for the actual mediation test. Otherwise, the mediation path is deleted. This procedure is repeated for all potential (eight) mediation paths of RQ2. However, as we can see from Fig. 2, FEMALE is not correlated with MASTERY-APPROACH, PERFORMANCE-AVOIDANCE and MASTERY-AVOIDANCE (that is, the subpath a in Fig. 1), and thereto Baron & Kenny (1986) can be satisfied only in the case of paths FEMALE = > HIGH EXPECTATIONS = >PERFORMANCE-APPROACH (Table 4. Panel A) or FEMALE = > LOW EXPECTATIONS = > PERFORMANCE-APPROACH (Table 4. Panel B). Concerning RQ3, there are in total 12 potential mediation paths. However, path a (please see Fig. 1) is significant only when FEMALE is a predictor of PERFORMANCE-APPROACH. Therefore, we present in the above Table three paths that have the possibility to satisfy Baron & Kenny (1986). Steps required for the mediation are: FEMALE => PERFORMANCE-APPROACH => EXAM POINTS (Panel C), FEMALE => PERFORMANCE-APPROACH = > TEAMWORK (Panel D), and FEMALE => PERFORMANCE-APPROACH => TOTAL POINTS (Panel E). The letters a, b, c, d and e denote the paths presented in Fig. 1.

(mediates) this association. When we use alternative modes of assessment, teamwork or total points as measures of course performance, we find no mediation effect. Consequently, our study contributes to the literature by showing that the role of the achievement goal as a mediator between gender and course performance depends on the mode of assessment.

To interpret our findings, we consider whether the mode of assessment (one component of the learning environment) favours certain qualities, attributes and knowledge, which on average benefit men more than women. Some prior studies have suggested that men prefer exams as an assessment method, whereas women prefer coursework (e.g., Furnham & Chamorro-Premuzic, 2005) and are better in collaborative learning (e.g., Curseu et al., 2018). Hence, we conjecture that the assessment of the course may have had some impact on our findings. Although various modes of assessment (exam and teamwork) were used in the course, somewhat more emphasis was placed on the course exam points (maximum 60/106 points), which may slightly favour those adopting the performance-approach goal. Collectively, the assessment and context features, competitiveness and the perception of accounting as a masculine field may constitute the underlying drivers of the gender differences in course performance. This is because the greater competitiveness of men may enhance their performance, leading to higher exam points; male students' performance is driven by the

J. Huikku et al.

desire to outperform others, especially in the masculine field of accounting.

From a practical point of view, our study sheds light on fundamental aspects of challenging and changing the learning environment of business schools' accounting courses. Any learning environment should be designed to be inclusive and not biased towards the learning orientation and expectations of any gender. While female students enter university with a somewhat higher level of prior knowledge, it is alarming that according to our results, they consider accounting a less attractive learning prospect than do their male counterparts. We conjecture that this is due to the perception of accounting as a masculine field. Haynes et al. (2017, p. 116) point out that given 'the masculine history of the [accounting] profession, stereotypes may be difficult to shift ... '. If stereotypes could be changed, women might be encouraged to compete in stereotypically male tasks (Niederle & Vesterlund, 2011). The first-year introductory accounting course can be pivotal in this regard. That is, it can encourage or discourage students from further study of accounting (Duff & Mladenovic, 2015, p. 325).¹⁷ Hence, accounting field. The lower exam performance of female students resulting from their less competitive goal-setting orientation is a worrying problem. It merits consideration in designing accounting education that should ensure all students equal chances to perform well.

Next, we will present potential remedies for the negative implications of gendered expectations and achievement goals in studying accounting, addressing both gender role stereotypes in accounting and gender differences in competitiveness. Strengthening support from peers and faculty is essential for gender inclusivity (Looze et al., 2018) and for mitigating competitive pressure particularly perceived by females (Niederle & Vesterlund, 2010). Peer tutoring, for example, may be effective in providing support for students by creating a realistic perception of accounting. Further, a more equal number of males and females in business schools may alleviate the gender gap in competitiveness because 'the gender composition of a group influences the individual's willingness to compete' (Niederle & Vesterlund, 2011, p. 622). Business schools could also implement practices to increase the number of female top-ranked faculty members to alleviate the gender imbalance (e.g., Treviño et al., 2017).

Considering the introductory accounting course learning environment, Mladenovic (2000) suggests that alignment (cf. Biggs, 1996) is the most effective approach in changing students' perceptions of accounting. Accordingly, improving the alignment of objectives, curriculum, teaching methods and assessments from the perspective of gender neutrality and inclusivity could successfully alleviate gender differences in expectations, goal setting and performance. Following the ideal of an aligned gender-neutral course environment, an assessment method with more emphasis on collaboration and group projects could be used to mitigate the consequences of gender differences. Gender neutrality can also be enhanced by communicating how both males and females are able to succeed in the professionals. As suggested by Niederle and Vesterlund (2011, p. 623), 'particularly successful may be institutions that simultaneously improve women's beliefs about their relative performance and increase the comfort they experience when competing'. In summary, there are various ways in which gender-specific aspects could be taken into consideration when accounting educators design their courses.

Our paper is subject to some limitations. First, we use biological gender as our independent variable, as is commonplace in the education literature. Nevertheless, it has been pointed out in the literature that gender is not necessarily binary or fixed. Hence, it would be fruitful to consider an individual's endorsement of masculine and feminine personality traits (e.g., Bebbington et al., 1997; Paver & Gammie, 2005). We agree with Severiens and Ten Dam (1997), who suggest that scholars could alternatively account for the gendered attributes and behaviours which the students have learned throughout their lifetimes by using constructed gender (i.e., gender identity) instead of just biological gender.

Second, we limit our study to students' achievement goals in an introductory accounting course. Our research could be extended by using our research design to cover other accounting courses. Future scholars could also extend our cross-sectional approach by addressing the dynamism of goal-orientations and expectations in longitudinal studies. Such dynamism may take many forms. For example, gender differences in risk-taking and competitiveness may diminish with experience, profession (Croson & Gneezy, 2009; Whiting & Wright, 2001) and age (Flory et al., 2018). Research has found that gender-specific achievement goal-orientation can change over time (e.g., D'Lima et al., 2014). Moreover, Ferreira and Santoso (2008) show that students' positive and negative perceptions of studying accounting may change to some extent during their studies. Finally, regarding course performance, both Doran et al. (1991) and Koh and Koh (1999) found that gender differences decrease over time, that is, during the later study years.¹⁹

Third, using data from a single country and a single university reduces the generalisability of the findings. However, investigating potential gender differences requires a research setting with a high level of gender equality in order to address the concern that major inequalities would bias the findings. Finland offers such a setting because among the Nordic countries, gender equality is at a higher level than in other countries (Looze et al., 2018). While we examine gender differences in achievement goals for an introductory accounting course in a largely gender-balanced setting, future research could examine this research question in a setting where

¹⁷ For example, Mladenovic (2000) examines how the design of the introductory accounting course could facilitate students' development of realistic perceptions of accounting.

¹⁸ Indeed, females show stronger performance than men in many areas of financial management. For example, prior studies have found associations between female CFOs and better bank loan prices (Francis et al., 2013), greater accounting conservatism (Francis et al., 2015), and fewer operations-related lawsuits in firms where women have more power in top management (Adhikari et al., 2019).

¹⁹ In general, although some studies do suggest gender differences in accounting course performance in the later study years (e.g., Bible et al., 2007; Gammie et al., 2003; Gracia & Jenkins, 2003; Mutchler et al., 1987; Surridge, 2009), several studies report no gender differences (e.g., Fallan & Opstad, 2014; Fogarty & Goldwater, 2010; Keef & Roush, 1997; Paver & Gammie, 2005; Tickell & Smyrnios, 2007).

J. Huikku et al.

equality is not as advanced as it is in the Nordic countries.

Fourth, scholars have suggested that individual-level social support systems related to support from family, student peers and schools may be essential in creating gender equality (Looze et al., 2018). Accordingly, future studies could extend our study by addressing how the various support systems influence students' goal orientation and course performance rather than gender per se.

Fifth, in our study, we focus on gender as a factor explaining achievement goals and expectations of learning. As a limitation, we acknowledge that our structural model could benefit from additional variables beyond our reach. Namely, studies have shown that achievement goals can be associated with personality (see e.g., Payne et al., 2007; Bibb et al., 2008), culture (e.g., in terms of race/ethnicity/nation of origin) (see Zusho & Clayton, 2011), upbringing (Singhal & Misra, 1994) and familial positioning (birth order/only-child influence) (Carette et al., 2011). In a similar vein, students' expectations of studying different disciplines may be related to personality (Pike, 2006), culture (Auyeung & Sands, 1997), upbringing (Tan & Laswad, 2006) and familial positioning (Morales, 1994).

Finally, although the sample size is sufficient for our analysis, we acknowledge that with a larger sample, statistical associations would potentially be stronger. However, we would like to point out that the sample of more than 1700 observations in the additional tests of course performance is clearly a large sample. Also, future qualitative research could provide further insights into gender differences in students' competitiveness and motivation for studying introductory accounting. Overall, the gender-driven differences documented in this study should be addressed in future research to identify potential reasons and remedies for them.

Acknowledgements

We would like to thank Patricia Everaert, Päivi Kinnunen, Tapani Kykkänen, Pia Lahti, Lauri Malmi, Jari Melgin, Martin Messner, Antti Miihkinen, and Susanne Weinberg-Krakowski for their constructive comments on earlier versions of this paper. In addition, we gratefully acknowledge comments from the participants of the Nordic Accounting Conference (2018, Copenhagen Business School) and EAA's Annual congress (2019, Paphos), and the research seminars at Aalto University School of Business (Helsinki), AUT University (Auckland), Brunel University London, Esade Business School (Barcelona), Stockholm School of Economics, University of Auckland, University of Eastern Finland (Kuopio), and University of Tampere. The financial support granted to the authors by the Helsinki School of Economics Support Foundation is acknowledged with gratitude.

Appendix

Variable definitions	
Variable:	
FEMALE	Equals 1 if the student is female, and zero if the student is male.
PERFORMANCE-	Latent CFA construct represented by the performance-approach goal items based on responses to the inventory AGQ-revised (Elliot
APPROACH	& Murayama, 2008).
MASTERY-APPROACH	Latent CFA construct represented by the mastery-approach goal items based on responses to the inventory AGQ-revised (Elliot &
	Murayama, 2008).
PERFORMANCE-	Latent CFA construct represented by the performance-avoidance goal items based on responses to the inventory AGQ-revised (Elliot
AVOIDANCE	& Murayama, 2008).
MASTERY-AVOIDANCE	Latent CFA construct represented by the mastery-avoidance goal items based on responses to the inventory AGQ-revised (Elliot &
	Murayama, 2008).
HIGH EXPECTATIONS	Latent CFA construct represented by the high expectations of learning accounting items (adopted from Duff & Mladenovic, 2015).
LOW EXPECTATIONS	Latent CFA construct represented by the low expectations of learning accounting items (adopted from Duff & Mladenovic, 2015).
EXAM POINTS	Exam points from the final exam of the introductory accounting course (max 60 points).
TEAMWORK	The total teamwork points (max 26 points).
TOTAL POINTS	The total course points (max 106 points).
PRIOR KNOWLEDGE	High school grade point average (max 40).
AGE	Student age (years).

References

Adcroft, A. (2011). The motivations to study and expectations of studying of undergraduate students in business and management. Journal of Further and Higher Education, 35(4), 521–543.

Auyeung, P., & Sands, J. (1997). Factors influencing accounting students' career choice: A cross-cultural validation study. Accounting Education, 6(1), 13–23. Baron, R. M., & Kenny, D. A. (1986). The moderator-mediator variable distinction in social psychological research: Conceptual, strategic, and statistical considerations. Journal of Personality and Social Psychology, 51(6), 1173–1182.

Adhikari, B. K., Agrawal, A., & Malm, J. (2019). Do women managers keep firms out of trouble? Evidence from corporate litigation and policies. Journal of Accounting and Economics, 67(1), 202–225.

Anderson, J. C., & Gerbing, D. W. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological Bulletin, 103*(3), 411–423.

Arthur, N., & Everaert, P. (2012). Gender and performance in accounting examinations: Exploring the impact of examination format. Accounting Education: International Journal, 21(5), 471–487.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

Bebbington, J., Thomson, I., & Wall, D. (1997). Accounting students and constructed gender: An exploration of gender in the context of accounting degree choices at two Scottish universities. *Journal of Accounting Education*, 15(2), 241–267.

Betts, J. R., & Morell, D. (1999). The determinants of undergraduate grade point average: The relative importance of family background, high school resources, and peer group effects. *Journal of Human Resources*, 34(2), 268–293.

Bibb, T., Steinmayr, R., & Spinath, B. (2008). Personality and achievement motivation: Relationship among big five domains and facet scales, achievement goals, and intelligence. *Personality and Individual Differences, 44*, 1454–1464.

Bible, L., Simkin, M. G., & Kuechler, W. L. (2007). Using multiple-choice tests to evaluate students' understanding of accounting. Accounting Education: International Journal, 17, 55–68.

Biggs, J. (1996). Enhancing teaching through constructive alignment. Higher Education, 32, 347-364.

Bouffard, T., Boisvert, J., Vezeau, C., & Larouche, C. (1995). The impact of goal orientation on selfregulation and performance among college students. British Journal of Educational Psychology, 65, 317–329.

Bråten, I., & Strømsø, H. I. (2006). Predicting achievement goals in two different academic contexts: A longitudinal study. Scandinavian Journal of Educational Research, 50(2), 127–148.

Breen, R., & Lindsay, R. (2002). Different disciplines require different motivations for student success. Research in Higher Education, 43(6), 693–725.

Brown, R. L. (1997). Assessing specific mediational effects in complex theoretical models. *Structural Equation Modeling*, *4*, 142–156.

Buckless, F. A., Lipe, M. G., & Ravenscroft, S. P. (1991). Do gender effects on accounting course performance persist after controlling for general academic aptitude? Issues in Accounting Education, 6(2), 248–261.

Carette, B., Anseel, F., & Van Yperen, N. W. (2011). Born to learn or born to win? Birth order effects on achievement goals. Journal of Research in Personality, 45(5), 500–503.

Carmona, S., & Ezzamel, M. (2016). Accounting and lived experience in the gendered workplace. Accounting, Organizations and Society, 49, 1-8.

Carpenter, V. L., Friar, S., & Lipe, M. G. (1993). Evidence on the performance of accounting students: Race, gender and expectations. *Issues in Accounting Education, 8* (1), 1–17.

Cavallo, A. M. L., Potter, W. H., & Rozman, M. (2004). Gender differences in learning constructs, shifts in learning constructs, and their relationship to course achievement in a structured inquiry, yearlong college physics course for life science majors. *School Science & Mathematics*, 104(6), 288–300.

Cook, D. A., Castillo, R. M., Gas, B., & Artino, A. R., jr. (2017). Measuring achievement goal motivation, mindsets and cognitive load: Validation of three instruments' scores. *Medical Education*, 51, 1061–1074.

Crawford, I., & Wang, Z. (2014). Why are first-year accounting studies inclusive? Accounting and Finance, 54, 419-439.

Croson, R., & Gneezy, U. (2009). Gender differences in preferences. Journal of Economic Literature, 47(2), 448-474.

Cui, Y. (2012). In N. J. Salkind (Ed.), Discriminant analysis in Encyclopedia of research design (Vol. 1). Sage.

Curşeu, P. L., Chappin, M. M. H., & Jansen, R. J. G. (2018). Gender diversity and motivation in collaborative learning groups: The mediating role of group discussion quality. Social Psychology of Education, 21, 289–302.

D'Lima, G. M., Winsler, A., & Kitsantas, A. (2014). Ethnic and gender differences in first-year college students' goal orientation, self-efficacy, and extrinsic and intrinsic motivation. The Journal of Educational Research, 107, 341–356.

Dong, Y., Girardone, C., & Kuo, J.-M. (2017). Governance, efficiency and risk taking in Chinese banking. The British Accounting Review, 49, 211-229.

Doran, B. M., Bouillon, M. L., & Smith, C. G. (1991). Determinants of student performance in accounting principles I and II. Issues in Accounting Education, 6(1), 74–84. Duff, A. (2004). Understanding academic performance and progression of first-year accounting and business economics undergraduates: The role of approaches to learning and prior academic achievement. Accounting Education, 13(4), 409–430.

Duff, A., & Mladenovic, R. (2015). Antecedents and consequences of accounting students' approaches to learning: A cluster analytic approach. The British Accounting Review, 47(3), 321–338.

Dull, R. B., Schleifer, L. L. F., & McMillan, J. J. (2015). Achievement goal theory: The relationship of accounting students' goal orientations with self-efficacy, anxiety, and achievement. Accounting Education: International Journal, 24(2), 152–174.

Dweck, C. (1986). Motivational processes affecting learning. American Psychologist, 41, 1040–1048.

Dweck, C. S. (1999). Self-theories: Their role in motivation, personality, and development. Philadelphia: Psychology Press.

Dweck, C., & Leggett, E. (1988). A social-cognitive approach to motivation and personality. Psychological Review, 95, 256-273.

Elliot, A. J., & Church, M. (1997). A hierarchical model of approach and avoidance achievement motivations. Journal of Personality and Social Psychology, 72, 218–232.

Elliot, A. J., & Harackiewicz, J. (1996). Approach and avoidance achievement goals and intrinsic motivation: A mediational analysis. *Journal of Personality and Social Psychology*, 72, 461–475.

Elliot, A. J., & McGregor, H. (2001). A 2 X 2 achievement goal framework. Journal of Personality and Social Psychology, 80(3), 501-519.

Elliot, A. J., & Murayama, K. (2008). On the measurement of achievement goals: Critique, illustration, and application. Journal of Educational Psychology, 100(3), 613–628.

Fallan, L., & Opstad, L. (2014). Beyond gender performance in accounting: Does personality distinction matter? Accounting Education: International Journal, 23(4), 343–361.

Ferreira, A., & Santoso, A. (2008). Do students' perceptions matter? A study of the effect of students' perceptions on academic performance. Accounting and Finance, 48, 209–231.

Flory, J. A., Gneezy, U., Leonard, K. L., & List, J. A. (2018). Gender, age, and competition: A disappearing gap? Journal of Economic Behavior & Organization, 150, 256–276.

Fogarty, T. J., & Goldwater, P. M. (2010). Beyond just desserts: The gendered nature of the connection between effort and achievement for accounting students. Journal of Accounting Education, 28, 1–12.

Fox, J. (1985). Effects analysis in structural equation models II: Calculation of specific indirect effects. Sociological Methods & Research, 14, 81–95.

Francis, B., Hasan, I., Park, J. C., & Wu, Q. (2015). Gender differences in financial reporting decision making: Evidence from accounting conservatism. Contemporary Accounting Research, 32(3), 1285–1318.

Francis, B., Hasan, I., & Wu, Q. (2013). The impact of CFO gender on bank loan contracting. Journal of Accounting, Auditing and Finance, 28(1), 53-78.

Furnham, A., & Chamorro-Premuzic, T. (2005). Individual differences and beliefs concerning preference for university assessment methods. *Journal of Applied Social Psychology*, 35(9), 1968–1994.

Furr, M. (2011). Scale construction and psychometrics for social and personality psychology. SAGE Publications Ltd.

Gammie, E., Paver, B., Gammie, B., & Duncan, F. (2003). Gender difference in accounting education: An undergraduate exploration. Accounting Education, 12(2), 177–196.

Gammie, E., & Whiting, R. (2013). Women accountants: Is the grass greener outside the profession? The British Accounting Review, 45, 83-98.

Gist, W. E., Goedde, H., & Ward, B. H. (1996). The influence of mathematical skills and other factors on minority student performance in principles of accounting. *Issues in Accounting Education*, 11(1), 49–60.

Gracia, L., & Jenkins, E. (2003). A quantitative exploration of student performance on an undergraduate accounting program of study. Accounting Education, 12(1), 15–32.

Gull, A. A., Nekhili, M., Nagati, H., & Chtioui, T. (2018). Beyond gender diversity: How specific attributes of female directors affect earnings management. The British Accounting Review, 50, 255–274.

Hall, N. C., Sampasivam, L., Muis, K. R., & Ranellucci, J. (2016). Achievement goals and emotions: The mediatinal roles of perceived progress, control, and value. British Journal of Educational Psychology, 86, 313–330.

Harackiewicz, J. M., Barron, K. E., Carter, S. M., Lehto, A. T., & Elliot, A. J. (1997). Predictors and consequences of achievement goals in the college classroom: Maintaining interest and making the grade. Journal of Personality and Social Psychology, 73(6), 1284–1295.

J. Huikku et al.

The British Accounting Review xxx (xxxx) xxx

Harackiewicz, J. M., Barron, K. E., & Elliot, A. J. (1998). Rethinking achievement goals: When are they adaptive for college students and why? *Educational Psychologist*, 33(1), 1–21.

Harackiewicz, J. M., Barron, K. E., Tauer, J. M., & Elliot, A. J. (2002). Predicting success in college: A longitudinal of achievement goals and ability measures as predictors of interest and performance from freshman year through graduation. *Journal of Educational Psychology*, *94*(3), 562–575.

Haynes, K. (2017). Accounting as gendering and gendered: A review of 25 years of critical accounting research on gender. Critical Perspectives on Accounting, 43, 110-124

Hijzen, D., Boekaerts, M., & Vedder, P. (2007). Exploring the links between students' engagement in cooperative learning, their goal preferences and appraisals of instructional conditions in the classroom. Learning and Instruction, 17(6), 673–687.

Hines, R. D. (1992). Accounting: Filling the negative space. Accounting, Organizations and Society, 17(3-4), 313-341.

Hulleman, C. S., Schrager, S. M., Bodmann, S. M., & Harackiewicz, J. M. (2010). A meta-analytic review of achievement goal measures: Different labels for the same constructs or different constructs with similar labels? *Psychological Bulletin*, 136(3), 422–449.

Keef, S. P., & Roush, M. L. (1997). New Zealand evidence on the performance of accounting students: Race, gender and self-concept. Issues in Accounting Education, 12 (2), 315–330.

Kelan, E. K., & Jones, R. D. (2010). Gender and the MBA. The Academy of Management Learning and Education, 9(1), 26-43.

Koh, M. Y., & Koh, H. C. (1999). The determinants of performance in an accountancy degree course. Accounting Education: International Journal, 8(1), 13–29.

Lipe, M. G. (1989). Further evidence on the performance of female versus male accounting students. Issues in Accounting Education, 4(1), 144–152.

Looze, M. E., Huijts, T., Stevens, G., Torsheim, T., & Vollebergh, W. (2018). The happiest Kids on earth. Gender equality and adolescent life satisfaction in Europe and North America. Youth Adolescence, 47, 1073–1085.

Lucas, U. (2000). Worlds apart: Students' experiences of learning introductory accounting. Critical Perspectives on Accounting, 11, 479-504.

Lucas, U., & Meyer, J. H. F. (2005). 'Towards a mapping of the student world': The identification of variation in students' conceptions of, and motivations to learn, introductory accounting. *The British Accounting Review, 37*, 177–204.

Lucas, U., & Tan, P. L. (2013). Developing a capacity to engage in critical reflection: Students' 'ways of knowing' within an undergraduate business and accounting programme. *Studies in Higher Education*, 38(1), 104–123.

Meece, J. L., Bower Glienke, B., & Burg, S. (2006). Gender and motivation. Journal of School Psychology, 44, 351–373.

Meyer, J. H. F., & Eley, M. G. (1999). The development of affective subscales to reflect variation in students' experiences of studying mathematics in higher education. *Higher Education*, 37, 197–216.

Meyer, J. H. F., & Shanahan, M. (2001). A triangulated approach to the modelling of learning outcomes in first year economics. *Higher Education Research and Development*, 20(2), 27, 145.

Mladenovic, R. (2000). An investigation into ways of challenging introductory accounting students' negative perceptions of accounting. Accounting Education, 9(2), 135–155.

Morales, C. A. (1994). Birth order theory: A case for cooperative learning. Journal of Instructional Psychology, 21(3), 246-249.

Mueller, R., & Hancock, G. (2008). Best practices in structural equation modeling. In J. Osborne (Ed.), Best practices in quantitative methods (pp. 488–508). Thousand Oaks, CA: SAGE Publications, Inc.

Murphy, R. (1982). Sex differences in objective test performance. British Journal of Educational Psychology, 52(2), 213-219.

Mutchler, J. F., Turner, J. H., & Williams, D. D. (1987). The performance of female versus male accounting students. Issues in Accounting Education, 2, 103–111.

Nicholls, J. (1984). Achievement motivation: Conceptions of ability, subjective experience, task choice, and performance. Psychological Review, 91, 328-346.

Niederle, M., & Vesterlund, L. (2007). Why women shy away from competition? Do men compete too much? Quarterly Journal of Economics, 122(3), 1067–1101.

Niederle, M., & Vesterlund, L. (2010). Explaining the gender gap in math test scores: The role of competition. *The Journal of Economic Perspectives*, 24(2), 129–144. Niederle, M., & Vesterlund, L. (2011). Gender and competition. *Annual Review of Economics*, 3, 601–630.

Nunnally, J. C. (1978). Psychometric theory. New York: McGraw-Hill.

Paver, B., & Gammie, E. (2005). Constructed gender, approach to learning and academic performance. Accounting Education: International Journal, 14(4), 427–444.
Payne, S. C., Youngcourt, S. S., & Beaubien, J. M. (2007). A meta-analytic examination of the goal orientation nomological net. Journal of Applied Psychology, 92(1), 128–150.

Pike, G. R. (2006). Students' personality types, intended majors, and college expectations: Further evidence concerning psychological and Sociological interpretations of Holland's theory. Research in Higher Education, 47, 801–822.

Powell, M., & Ansic, D. (1997). Gender differences in risk behaviour in financial decision-making: An experimental analysis. Journal of Economic Psychology, 18, 605–628.

Preacher, K. J., & Hayes, A. F. (2008). Asymptotic and resampling strategies for assessing and comparing indirect effects in multiple mediator models. *Behavior Research Methods*, 40(3), 879–891.

Rankin, M., Silvester, M., Vallely, M., & Wyatt, A. (2003). An analysis of the implications of diversity for students' first level accounting performance. Accounting and Finance, 43, 365–393.

Schreiber, J. B., Nora, A., Stage, F. K., Barlow, E. A., & King, J. (2006). Reporting structural equation modelling and confirmatory factor analysis results: A review. The Journal of Educational Research, 99(6), 323–338.

Senko, C., Hulleman, C. S., & Harackiewicz, J. M. (2011). Achievement goal theory at the crossroads: Old controversies, current challenges, and new directions. *Educational Psychologist*, 46(1), 26–47.

Severiens, S., & Ten Dam, G. (1997). Gender and gender identity differences in learning styles. *Educational Psychology*, *17*(1–2), 79–93.

Singhal, R., & Misra, G. (1994). Achievement goals: A situational-contextual analysis. International Journal of Intercultural Relations, 18(2), 239-258.

Struyven, K., Dochy, F., & Janssens, S. (2005). Students' perceptions about evaluation and assessment in higher education: A review. Assessment & Evaluation in Higher Education, 30(4), 325–334.

Surridge, I. (2009). Accounting and finance degrees: Is the academic performance of placement students better? Accounting Education: International Journal, 18, 471–485.

Tan, L. M., & Laswad, F. (2006). Students' beliefs, attitudes and intentions to major in accounting. Accounting Education, 15(2), 167–187.

Tempelaar, D. T., Rienties, B., Giesbers, B., & Gijselaers, W. H. (2015). The pivotal role of effort beliefs in mediating implicit theories of intelligence and achievement goals and academic motivations. Social Psychology of Education, 8, 101–120.

Tho, L. M. (1994). Some evidence on the determinants of student performance in the university of Malaya introductory accounting course. Accounting Education, 3(4), 331–340.

Tickell, G., & Smyrnios, K. X. (2007). Predictors of tertiary accounting students' academic performance: A comparison of year 12-to-university students with TAFE-touniversity students. Journal of Higher Education Policy and Management, 27(2), 239–259.

Treviño, L. J., Balkin, D. B., & Gomez-Mejia, L. R. (2017). How "doing gender" leads to gender imbalances in the higher ranks in colleges of business [and how to "Undo Gender"]. Academy of Management Learning & Education, 16(3), 439-453.

Turner, G., & Gibbs, G. (2010). Are assessment environments gendered? An analysis of the learning responses of male and female students to different assessment environments. Assessment & Evaluation in Higher Education, 35(6), 687–698.

Tyson, T. (1989). Grade performance in introductory accounting courses: Why female students outperform males. Issues in Accounting Education, 4(1), 153-160.

Walstad, W., & Robson, D. (1997). Differential item functioning and male-female differences on multiple-choice tests in economics. *The Journal of Economic Education*, 28(2), 155–171.

Whiting, R. H., & Wright, C. (2001). Explaining gender inequity in the New Zealand accounting profession. *The British Accounting Review, 33*(2), 191–222.

Woodfield, R., Earl-Novell, S., & Solomon, L. (2005). Gender and mode of assessment at university: Should we assume female students are better suited to coursework and males to unseen examinations? Assessment & Evaluation in Higher Education, 30(1), 35–50.

Zusho, A., & Clayton, K. (2011). Culturalizing achievement goal theory and research. Educational Psychologist, 46(4), 239-260.